Author's response to reviews

Title: Endoscopic Inside stent placement is suitable as a bridging treatment for preoperative biliary tract cancer

Authors:

Noritoshi Kobayashi (norikoba@yokohama-cu.ac.jp)
Seitaro Watanabe (smaeda@yokohama-cu.ac.jp)
Kunihiro Hosono (hiro1017@fukuhp.yokohama-cu.ac.jp)
Kensuke Kubota (kubotak@yokohama-cu.ac.jp)
Atsushi Nakajima (nakajima-tky@umin.ac.jp)
Takashi Kaneko (taku47@urahp.yokohama-cu.ac.jp)
Kazuya Sugimori (sugimori@urahp.yokohama-cu.ac.jp)
Motohiko Tokuhisa (tokuhsa@yokohama-cu.ac.jp)
Ayumu Goto (qq9v6m3u9@able.ocn.ne.jp)
Ryutaro Mori (rmori@ck2.so-net.ne.jp)
Koichi Taniguchi (t1006koich@yahoo.co.jp)
Ryusei Matsuyma (ryusei@terra.dti.ne.jp)
Itaru Endo (endoi@med.yokohama-cu.ac.jp)
Shin Maeda (smaeda@yokohama-cu.ac.jp)
Yasushi Ichikawa (yasu0514@med.yokohama-cu.ac.jp)

Version: 6 Date: 8 January 2015

Author's response to reviews:

Editorial Office
January 3rd, 2015

Dear Editors

We are pleased to review our manuscript entitled “Endoscopic Inside stent placement is suitable as a bridging treatment for preoperative biliary tract cancer” for publication consideration in BMC Gastroenterology. I revised this article according to the editor’s and reviewers’ comments.

We added the “In 2012, Moon reported on the use of a modified fully covered self-expandable metal stent 14). The metallic stents were put in place for the temporary treatment of benign biliary strictures above the ampulla of Vater and removed after 3 to 5 months. Technical success and improvement in ductal strictures were achieved for almost all the cases, and stent-induced bile duct injuries were not observed.” in the part of discussion and reference 14 according to the referee 1’s comment.

We asked a native English speaking colleague to help you copyedit the paper. I attached certification of English editing.

I included all Tables in the main manuscript file, after references.
I included section: Competing interests, and moved "The authors declare that they have no competing interests" sentence there.
I clarified Authors Contributions.

We believe that our paper would be of great interest to the readers of BMC Gastroenterology and look forward to hearing your response.

Sincerely yours,
Noritoshi Kobayashi, M.D., Ph.D.
Medical Oncology Division, Yokohama City University School of Medicine
Gastroenterology Division, Yokohama City University School of Medicine
3-9, Fuku-ura, Kanazawa-ku, Yokohama, 236-0004, Japan.
TEL: +81-45-787-2640, FAX: +80-45-784-3546
E-mail: norikoba@yokohama-cu.ac.jp